Author's response to reviews

Title: When battery exhaustion lets the lame walk: A case report on the importance of long-term stimulator monitoring in deep brain stimulation

Authors:

Martin Sommer (msommer@gwdg.de)
Elisabeth M Stiksrud (e.stiksrud@gmx.de)
Kajetan von Eckardstein (kajetan.voneckardstein@med.uni-goettingen.de)
Veit Rohde (veit.rohde@med.uni-goettingen.de)
Walter Paulus (wpaulus@med.uni-goettingen.de)

Version: 5
Date: 2 June 2015

Author's response to reviews: see over
Dear Prof Fabrizio Stocchi, dear Mr Jhonell De Los Santos

Thank you for accepting the manuscript. In the updated version that we herewith submit, we made the changes you requested in your acceptance mail dated June 1, 2015.

1. We listed the email addresses of all authors on the title page.

2. We changed dysarthrophonic on line 57 to dysarthric, which seems to be the proper adjective according to an internet search.

3. There was no funding information on the title page. We shifted the one at the end of the text to the acknowledgement section.

Thank you for your careful evaluation of this manuscript.

Yours sincerely,

Martin Sommer, M.D